Author's response to reviews

Title: A Realist Review of Interventions and Strategies to promote evidence-informed health care: A Focus on Change Agency

Authors:

brendan g mccormack (bg.mccormack@ulster.ac.uk)
joanne rycroft-malone (j.rycroft-malone@bangor.ac.uk)
kara deKorby (decorbk@mcmaster.ca)
Alison Hutchinson (alison.hutchinson@deakin.edu.au)
Tracey Bucknall (tracey.bucknall@deakin.edu.au)
Bridie Kent (brigid.kent@plymouth.ac.uk)
Alyce Schultz (alyceaschultz@gmail.com)
Erna Snelgrove-clarke (erna.snelgrove-clarke@dal.ca)
Cheyl stetler (cheryl.stetler@comcast.net)
Marita titler (mtitler@umich.edu)
Lars Wallin (lars.wallin@ki.se)
Valerie Wilson (ValerieW@chw.edu.au)

Version: 6 Date: 21 August 2013

Author's response to reviews:

To the editorial team.

I have accepted the changes made by copyeditors to the final manuscript. Two suggested changes have been rejected:

1 - the editors suggested adding in 'street addresses' to each author’s affiliations. This is not necessary and the addresses supplied are sufficient.

2 - Page 18: 'Outcome' - the sentence needs to remain as is in the text.